Author’s response to reviews

Title: Modern contraceptive use among sexually active men in Uganda: Does discussion with a health worker matter?

Authors:

Allen Kabagenyi Ms. (allenka79@yahoo.com)
Patricia Ndugga Ms. (patnduggab@yahoo.com)
Stephen Ojambo Wandera Mr. (swandera@gmail.com)
Betty Kwagala Dr. (kkwagala@gmail.com)

Version: 2 Date: 14 May 2013

Author’s response to reviews:

Dear Editor,

Thank you for accepting our manuscript and for permitting us to make some changes as suggested.

We have revised the manuscript and made modifications in the methodology section, that the data was obtained from DHS website accordingly.

Attached is a document explaining the other details.

Thank you.

Allen k